EE 201 CIRCUIT THEORY

Fall 2000

Information Sheet

Instructor : Ömer Morgül Office/Phone : 203/1529 Office Hours : Tuesday 15:30, 17:30.

TA’s : H. Köroğlu, I. Abdel-Hafez, T. Aytaç E. D. Kavak, O. Yücesan, E. Tomur.

Class Time : Section 1 Monday 9:40-10:30 Section 2 Monday 10:40-12:30 Wednesday 10:40-12:30 Wednesday 13:40-14:30 Lab Hours : Friday 9:40-12:30 Monday 13:40-16:30 Classroom : BZ-03 BZ-03

Text :

Electric Circuits, N. Balabanian, McGraw-Hill, 1994. (Required).

Linear and Nonlinear Circuits , L. O. Chua, C. A. Desoer and E. S. Kuh, McGraw-Hill, 1987. (Highly Recommended).

Additional References :

Basic Circuit Theory, C. A. Desoer and E. S. Kuh, McGraw-Hill, 1969,(15th prt. 1985) Analysis and Design of Linear Circuits R. E. Thomas, A. J. Rosa, Prentice-Hall, 1994. Linear Circuits, M. E. Van Valkenburg, B. K Kinariwala, Prentice-Hall, 1982. Applied Circuit Theory, P. R. Adby, Ellis-Horwood, 1980.

Microelectronic Circuits, A. S. Sedra and K. C. Smith, Holt, Rinehart and Winston, 1987. Circuits, Signals and Systems, W. McC. Siebert, MIT Press, 1986. Engineering Circuit Analysis, W. H. Hayt and J. E. Kemmerly, McGraw-Hill, 1986. Basic Engineering Circuit Analysis, J. D. Irwin, MacMillan, 1987.

You can get additional references in the library. The related books are usually indexed by the call number TK454...

Grading : Homeworks 15% Midterm 1 20% Midterm 2 20% Lab 15% Final 30%

Exam Schedule : Midterm 1 Friday, October 27 Midterm 2 Friday, November 24 Final Not scheduled yet.

Homeworks : HW’s will be assigned weekly on every Wednesday, and will be due to every following Wednesday, no later than 5 p.m., to the instructor’s mail box at room 218. Late HW’s will not be graded.